Title: Heme Iron Polypeptide For The Treatment Of Iron Deficiency Anemia In Non-Dialysis Chronic Kidney Disease Patients: A Randomized Controlled Trial

Authors:

Shankar P Nagaraju (shankarmmcmed@yahoo.com)
Adam Cohn (a_cohn@yahoo.com)
Ayub Akbari (aakbari@toh.on.ca)
Janet L Davis (jadavis@toh.on.ca)
Deborah L Zimmerman (dzimmerman@ottawahospital.on.ca)

Version: 3 Date: 19 February 2013

Author's response to reviews: see over
Hayley Henderson  
Executive Editor  
Biomed Central  
Floor 6,  
236 Gray’s Inn Road  
London, WC1X 8HL  
UK

Dear Dr Henderson,

Thank-you for the opportunity to re-submit a revised version of our manuscript to BMC Nephrology. The only concerns that remained were issues about p-values in Table 1 and p-values that highlight the changes from baseline to 6 months in both groups. We have added p-values to Table 1 as requested and we have added a Table 2 to highlight the changes from baseline to 6 months. We have removed the values from the text as we felt it was redundant to have them in both places.

Yours truly,

Deborah Zimmerman MD, MSc  
Director Home Hemodialysis  
Ottawa Hospital, Ottawa ON